Author's response to reviews

Title: Prevalence And Modes Of Complementary And Alternative Medicine Use Among Peasant Farmers With Musculoskeletal Pain In A Rural Community In South-Western Nigeria

Authors:

Chidozie E Mbada (doziembada@yahoo.com)
Tijani L Adeyemi (lukmanadeyemi@yahoo.com)
Rufus A Adedoyin (radedoyi@yahoo.com)
Hakeem D Badmus (hbadmus@yahoo.com)
Taofeek O Awotidebe (tidebet@yahoo.com)
Olujide O Arije (olujide_arije@yahoo.com)
Olorunfemi S Omotosho (davfemsinc@yahoo.com)

Version: 5 Date: 14 April 2015

Author's response to reviews: see over
The Editor in Chief,
Journal: BMC Complementary and Alternative Medicine

Dear Sir,

Correction to Revised Manuscript - Prevalence And Modes Of Complementary And Alternative Medicine Use Among Peasant Farmers With Musculoskeletal Pain In A Rural Community In South-Western Nigeria

Thank you for the further valuable comments of the reviewer on our revised manuscript. I hereby state that all the comments were considered and corrections were implemented. Specifically, the comments to each comment is presented in the table below:

<table>
<thead>
<tr>
<th>SN</th>
<th>Comment</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>I don't agree with the definition of CAM used in this study as it is confusing for readers to understand CAM ‘were not SUGGESTED by a HEALTH CARE PROVIDER’. Also, it is not clear to use ‘frequency of CAM use’ and ‘CAM provider use’ as sub-titles in Table 3. What does ‘frequency of CAM use’ stand for?</td>
<td>- The phrase ‘were not SUGGESTED by a HEALTH CARE PROVIDER’ was expunged from the new revised manuscript and also in the Questionnaire. - The sub-title ‘frequency of CAM use’ was recast as “Frequency of use of CAM”. The sub-title was meant to present the result on the number of people that use CAM for MSP on daily, weekly, monthly etc. basis. - CAM provider use’ was recast as “Frequency of visits to CAM Practitioners”.</td>
</tr>
<tr>
<td>2</td>
<td>Please change another word ‘mind-body therapy’ in Table 2 for consistency.</td>
<td>This word ‘mind-body therapy’ was changed to ‘mind-body intervention’ in table 2 for consistency.</td>
</tr>
<tr>
<td>3</td>
<td>Need language corrections regarding ‘CAM use’ and ‘CAM use for musculoskeletal disorders’.</td>
<td>The required language corrections were effected in the discussion section.</td>
</tr>
</tbody>
</table>

All the corrections made to the manuscript were presented in bold format.

Thank you immensely,

Chidozie E. Mbada PhD.